EO-1 Weekly Status Week of August 17 – August 23, 2007

Day of Year 229 - 235

Mission Day 2459 - 2465

Earth Observing-One (EO-1) General:

There were 101 Data Collection Events (DCEs) scheduled this week.  

Instruments underwent decontamination cycling as follows:

The Hyperion hyperspectral imager (HSI) performed a deicing from 08/17/2007 at 03:54z (07-239) to 08/17/2007 at 17:40z (07-239).  

The Advanced Land Imager (ALI) performed an outgassing from 08/17/2007 at 04:02z (07-239) to 08/17/2007 at 17:54z (07-239).  

EO-1 Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) telemetry playback anomaly reported previously.  

Instruments

All instruments operated nominally this week.  
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Technology Activities:
In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

Sensor Web & Virtual Observatory Demonstrations

Another teleconference was held between the Wildfire Research and Applications Program (WRAP) and the Advanced Information System Technology (AIST) Sensor Web collaborators on Thursday, 23 August.  The following key items of interest were discussed:

· Discussion occurred on whether all parties are using the same data sources for target computation.  It was concluded that Draper would only send targets found in the GeoBliki WFS for MODIS hot pixels.
· It was agreed that a key objective is to obtain contemporaneous imagery produced by the EO-1, Unmanned Aerial System (UAS), and ASTER for this upcoming demonstration.

· It was also agreed that on Monday, 27 August key parties would have a telecon to coordinate activities leading up to next week’s UAS flight.

· The Thursday, 23 August UAS WRAP flight was cancelled and rescheduled for Friday, 24 August.

· The Earth Science Technology Office (ESTO) AIST annual review was held for the Interoperable Sensor Architecture for Sensor Web effort.  The reviewers requested a better linkage between the original task objectives and the accomplishments to date.  The review package is being modified to respond to this action.

· Figure 1 below depicts the mashup for the Zaca fire produced on August 16 during the first WRAP demonstration flight.  The UAS flight path is shown as blue lines.  The hot pixels detected by the MODIS Rapidfire workstation are shown as yellow and orange dots.  The green overlay was produced by the onboard UAS instrument.  The base map is provided by Google Earth and the lower right thumbnail is an image of the entire Zaca fire with the UAS flight path superimposed.
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· In support of the Global Earth Observation System of Systems (GEOSS) Pilot, EO-1 sensor web components are being updated to autonomously trigger EO-1 for South Africa fire targets that are closest to power lines.  The MERAKA Institute in South Africa has set up several WFS’s and has developed power line specification files in KML format.  The target timeframe to do this experiment is within a month if possible.
Figure 2 below contains the mashup of power line and MODIS hot pixels overlaid in Google Earth map.
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· We are experimenting on how to generate EO-1 ground track and swath information by exporting KML files from STK using current ephemeris data.  The KML files will show the  EO-1 position on a Google Earth map.
Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 all week.

Ground System

Hardware Status

	Hardware
	Status
	Comments

	Primary R/T System
	
	Green
	

	Backup R/T System
	
	Green
	Backup R/T system will be revisited.  Red-Hat OS will be upgraded from version 3 to version 4. ASIST will be re-installed and FOT will conduct regression testing to verify the compatibility of the new operating system.  Once Red-Hat 4 and the latest version of ASIST are successfully tested the Primary system will be upgraded as well. 

	bMOC at DataLynx
	
	Green
	

	Playback System
	
	Green
	

	Mission Planning
	
	Yellow
	EO-1s DBA is scheduled to setup the mission planning Oracle database on Axle after the security audit.  Mission planners will continue to perform all tasks using Speedwheel and Bullseye.

	Flight Dynamics
	
	Green
	XFDS (“Auto-Products”) has been installed on one of the new flight dynamics machines.  Parallel testing is in progress.

	DTAS Server
	
	Yellow
	As the result of a preliminary security scan, a part of the DTAS trending system code has been flagged as a high-risk element.  The server is currently off of the network and will be populated with data manually by the FOT using external data storage units.  FOT and the system administrators are discussing the options of how to keep the server on the network.

	ANS
	
	Green
	

	Flight Software Lab
	
	Green
	

	Admin System
	
	Green
	


Operations

New Discrepancy Reports (DR)

There are no new requests this week.

New Enhancement Requests (ER)
There are no new requests this week  

New Operations Database Change Requests (DBCR)

There are no new requests this week.

Ground and Space Network

EOS Data and Operations System (EDOS), GSFC, MD, USA

There are no major problems to report.
McMurdo Ground Station (MGS), McMurdo, Antarctica

There are no major problems to report.

Wallops Ground Station (WGS), Wallops Island, VA, USA

The 11-Meter Antenna status is Red through Monday August 27, 2007.  The GNSTAT report indicates the following as the problem:  SCC Upgrade, Receiver Upgrade (enertec installation) and Auxiliary Shelter Move.

DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.

Alaska Ground Station (AGS), Poker Flat, AK, USA 

There are no major problems to report.

Svalbard Ground Station (SGS), Longyearbyen, Norway

There are no major problems to report.

Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.
USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA
LGS is now being scheduled only for contingencies.
Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

Australian Centre for Remote Sensing reports that TERSS has been having technical issues.  The antenna will be down until further notice.  Original estimates listed a return to operations in early July.  The FOT requested an update on the status of the HGS repairs.  The latest report from Hobart is that the next update on the status of the repairs is scheduled for mid-August. 

Cordoba Ground Station (CGS), Cordoba, Argentina

No passes were scheduled this week. 

Upcoming Events:

· Lunar Calibration scheduled for August 28, 2007

Imagery Status:

Scenes and Engineering Cals planned for week of Aug 17 – Aug 23, 2007                          101

Total scenes and engineering calibrations planned for entire mission – approximately 35,731

Total Scenes:  ALI scenes in the Level 0 archive            32,325 (as of Aug 23, 2007)

                        Hyperion scenes in the Level 0 archive    32,071
Publications and Presentations Status (as of 07-05-07):

404 publications 

284 external presentations

51 articles and press releases

Active Fire Image from Zaca Fire


(Santa Barbara County, California)


collected 08-16-2007, 21:46:28 Z


bands:


10.26 - 11.26 µm (VIIRS M15)


1.55 -1.75 µm (ETM+ 5)


3.60 - 3.79 µm (VIIRS M12)
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